Author’s response to reviews

Title: The Analgesic Efficacy and Safety of Peri-articular Injection Versus Intra-articular Injection in One-stage Bilateral Total Knee Arthroplasty: A Randomized Controlled Trial

Authors:

Kai-Yuan Cheng (ckymedicine@163.com)
Bin Feng (pumcfeng@163.com)
Hui-Ming Peng (penghuiming@139.com)
Yan-Yan Bian (bianyan4@163.com)
Lin-Jie Zhang (linjie0222@qq.com)
Chang Han (hanchanglc@gmail.com)
Gui-Xing Qiu (qiuguixingpumch@gmail.com)
xisheng weng (xshweng@medmail.com.cn)

Version: 2 Date: 11 Dec 2019

Author’s response to reviews:

Cover letter

Dear Martin Kaczocha,

Thank you for reviewing the manuscript “The analgesic efficacy and safety of peri-articular injection versus intra-articular injection in one-stage bilateral total knee arthroplasty: a randomized controlled trial (BANE-D-19-00701R1)”. I appreciate it very much for the rapid review of our revised manuscript and acceptance of the major revisions. On the other hand, we have corrected the abbreviations of the journal title of ref. 17, 18 to 24 and 30. The revised portions were marked in green highlight.

I am looking forward to receiving your decision and comments. If you have any further question, don’t hesitate to contact us.

Best wishes for the upcoming Christmas’s Day!

Sincerely,
Dr Xisheng Weng